Author’s response to reviews

Title: Digenetic inheritance of SLC12A3 and CLCNKB genes in a Chinese girl with Gitelman syndrome

Authors:

yuan mei Kong (kongyuanmei@163.com)
Ke Xu (xuke@zhuanhuayixue.org)
Ke Yuan (21618001@zju.edu.cn)
Jianfang Zhu (6507032@zju.edu.cn)
Weiyue Gu (guweiyue@zhuanhuayixue.org)
Li Liang (zdliangli@zju.edu.cn)
Chunlin Wang (hwangcl@zju.edu.cn)

Version: 2 Date: 26 Mar 2019

Author’s response to reviews:

The manuscript, Digenetic inheritance of SLC12A3 and CLCNKB genes in a Chinese girl with Gitelman syndrome, was edited for proper English language, grammar, punctuation, spelling, and overall style by highly qualified native English speaking and professional scientific editors.